Author's response to reviews

Title: Endorsement for improving the quality of reports on randomized controlled trials of traditional medicine journals in Korea: A systematic review

Authors:

- Jiae Choi (choijiae0723@gmail.com)
- Ji Hee Jun (zhixi04@kiom.re.kr)
- Kun Hyung Kim (pdchrist@gmail.com)
- Myeong Soo Lee (drmslee@gmail.com)
- Byung Kab Kang (bkkang@kiom.re.kr)

Version: 5 Date: 14 October 2014

Author's response to reviews: see over
MS: 1165012195133335

Title: Endorsement for improving the quality of reports on randomized controlled trials of traditional medicine journals in Korea: A systematic review

Jiae Choi, Ji Hee Jun, Kun Hyung Kim, Byung Kab Kang and Myeong Soo Lee

Dear Dr. Zhaoxiang Bian,

On the behalf of my co-author, I would like to thank you for arranging peer-review of our manuscript and for your invitation to submit a revised version. We appreciate the effort of the reviewers, and believe that their constructive suggestions have resulted in a stronger manuscript for the Trials readers.

We thank you in advance for your consideration of this manuscript.

Yours faithfully,

Myeong Soo Lee, PhD.